
SPS status

• Physics started on Monday 19:00 (instead of 8:00). Problems with 
beam loss monitors during weekend and Monday during the daytime.

• Tuesday scheduled stop for interventions in NA from 9:00 to 11:00. 
Beam only back around 14:30 because of RF (broken module)

• BLM problem still under investigation (general in all transfer lines). 
For TT20 the problem is under control but not yet completely 
operational.


